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•
 

FAR DETECTOR:
•

 

99.8% live time 
•

 

More spring snow, blizzards.  All state facilities shut down on 
Friday/Saturday, including the Soudan Lab 
•But experiments run happily along, no power or network troubles

•
 

NEAR DETECTOR:
•

 

Up ~98% of the time
•

 

Detector B-Field reversed Tuesday, 8 April
•Will do this regularly (~ each 2 weeks) from now on, to accumulate 
good statistics in our quest to reduce systematic errors from bfield, 
tracking, and alignment
•Done previously once for a week of beam data, plus 4 weeks of 
cosmics

 

data, to allow cosmic ray charge ratio measurement
•

 

Electrical work allowed investigations of potential light-leaks 
with a Dark Test, valuable debugging information gathered
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•
 

No NuMI
 

specific problems this week
•

 
Opportunistic maintenance includes:

Cleaned condensate drain line for dehumidifier A unit 
o

 

Done previously for B unit
Installed remote monitoring for Unit A dehumidifier heaters
Repaired MI65 condensate neutralization tank level sensor
Repaired chilled water pump “A”
Remote control & monitoring installed for target pile chiller
Rerouted three sets of outlets in Pre-Target to put out of 
harms way in event of double sump pump failure
Repaired small external water leak for Q120 manifold 
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•
 

Protons/pulse:
~23x1012: 2+9 6-turns 
mixed

o

 

Dominates this week, 
dialed back to 6 turns 
during initial BP02 
troubles

~26x1012: 2+9 7-turns 
mixed
~32x1012: NuMI only 
(during pbar recycler 
transfers)

<2.46 E13 ppp>
Best week 2.59 E13
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~190kW with 6 turns, 
~230kW with 7 turns, 
~270kW in NuMI only 
(no MTEST this week)

No NuMI specific downtime

<192 kW>
Best week 214 kW

0.85E18
Best week 6E18 pot
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end of 2007 shutdown
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Weekly Avg. needed to meet 2E20 pot by 30 September (no shutdown!)
Same, but with 4-weeks float
7-day running average
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• In the more traditional but more pessimistic looking view:
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end of 2007 
shutdown

4.403 E 20 total POT to NuMI
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